Author's response to reviews

Title: Risk Factors for treatment delay in pulmonary tuberculosis in Recife, Brazil

Authors:

Martinho APS dos Santos (martinhoapsantos@hotmail.com)
Maria FPM Albuquerque (militao@cpgam.fiocruz.br)
Ricardo AA Ximenes (rximenes@elogica.com.br)
Norma LCL Lucena-Silva (nlucena@cpgam.fiocruz.br)
Cynthia C Braga (braga@cpgam.fiocruz.br)
Antonio RL Campelo (rcampelo@nlink.com.br)
Odimariles MS Dantas (odimariles@globo.com)
Ulisses R Montarroyos (ulisses@cpgam.fiocruz.br)
Wayner V Souza (wayner@cpgam.fiocruz.br)
Alexandre M Kawasaki (kawasaki@hotlink.com.br)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)

Version: 6 Date: 11 March 2005

Author's response to reviews:

Dear Dr. Borthwick

Thank you very much for your attention. We are glad to know that our revised manuscript was accepted. Following the instructions, the formatting changes were performed as well as minor changes to the tables (the total percentages were omitted).

We will be available for further directions.

Cordially,

Cynthia Braga